Author's response to reviews


Authors:

Rachel Eyre (Rachel.Eyre@ncl.ac.uk)
Richard G Feltbower (R.G.Feltbower@ncl.ac.uk)
Peter W James (P.W.James@ncl.ac.uk)
Karen Blakey (Karen.Blakey@ncl.ac.uk)
Emmanuel Mubwandarikwa (e.mubwandarikwa@tiscali.co.uk)
David Forman (d.forman@leeds.ac.uk)
Patricia A McKinney (P.A.McKinney@leeds.ac.uk)
Mark S Pearce (M.S.Pearce@ncl.ac.uk)
Richard JQ McNally (Richard.McNally@ncl.ac.uk)

Version: 8 Date: 25 June 2010

Author's response to reviews: see over
POINT-BY-POINT RESPONSES TO FORMATTING CHANGES

We have corrected some minor typing errors and thoroughly proof read the manuscript.

Major revisions

- **Authors’ contributions:** We now indicate that all authors have read and approved the final manuscript.

Minor revisions

- **Highlighting and underlining:** We have removed the highlighting and underlining.
- **Figures:** No changes have been made.
- **Typography:** Final corrections and revisions have been made.